Author's response to reviews

Title: Petiveria alliacea extracts uses multiple mechanisms to inhibit growth of human and mouse tumoral cells.

Authors:

Claudia C Urueña (curuena@javeriana.edu.co)
Claudia C Cifuentes (ccifuent@javeriana.edu.co)
Diana D Castañeda (d.castaneda@javeriana.edu.co)
Amparo A Arango (amarco_02@yahoo.com)
Punit P Kaur (pkaur@medicine.tamhsc.edu)
Alexzander A Asea (asea@medicine.tamhsc.edu)
Susana S Fiorentino (susana.fiorentino@javeriana.edu.co)

Version: 2 Date: 3 June 2008

Author's response to reviews:

Dear Dr. Norton,

Concerning the revision of the manuscript “Petiveria alliacea extracts uses multiple mechanisms to inhibit growth of human and mouse tumoral cells”, we will like complete Table 1A and Table 1B because there are missing parts. Therefore, we are resending the whole Table 1 (A and B). Also we are adding a paragraph in Materials and Methods in plant extract preparation related to the identification of the plant and a Reference that was incomplete.